

# Sovereign Default and Liquidity: the Case for a World Safe Asset

François Le Grand, Xavier Ragot

# ▶ To cite this version:

François Le Grand, Xavier Ragot. Sovereign Default and Liquidity: the Case for a World Safe Asset. 2017. hal-03471758

# HAL Id: hal-03471758 https://sciencespo.hal.science/hal-03471758

Preprint submitted on 9 Dec 2021

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

**Discussion Paper** 



# SOVEREIGN DEFAULT AND LIQUIDITY: THE CASE FOR A WORLD SAFE ASSET

François Le Grand and Xavier Ragot

SCIENCES PO ECONOMICS DISCUSSION PAPER

No. 2017-04

# Sovereign Default and Liquidity: The Case for a World Safe Asset\*

François Le Grand

Xavier Ragot<sup>†</sup>

October 2017

#### Abstract

We present a general equilibrium model of the world economy where sovereigns face idiosyncratic risks and can default on their debt. In this model, the world interest rate is determined through the global financial market equilibrium, the amount of safe asset is endogenous and determines international risk sharing. Non-trivial multiple equilibria naturally arise, due to the endogeneity of the interest rate. These equilibria can be ranked according to their aggregate welfare, such that equilibria with a higher quantity of safe assets correspond to higher welfare. Due to a shortage in the safe asset supply, even the equilibrium with the highest welfare is not constrained-efficient. Finally, we prove that a world fund issuing a safe asset can reach the constrained-efficient aggregate welfare. With a standard calibration, the size of the fund is found to be around 4.1% of the world GDP. Its relationship with the Special Drawing Rights of the IMF is discussed.

# 1 Introduction

The global economy experiences two related features. The first one is the pervasiveness of sovereign default. The largest default in history (by present value) was the 2012 Greek restructuring that covered more than  $\notin$  200 bn of privately held debt (Tomz and Wright 2013). Debates about a larger restructuring of the Greek debt or about the default of another developed country indicate that sovereign default may reach another order of magnitude in the near future. Even

<sup>&</sup>lt;sup>\*</sup>We would like to thank Klaus Adam, Edouard Challe, Emmanuel Farhi, Jonathan Heathcote, Hugo Hopenhayn, Vincenzo Quadrini, Pierre-Olivier Weill, and seminar participants at ETH Zurich, European Economic Association meeting, Federal Reserve Bank of Minneapolis, ENSAE-Crest, NHH-UiO workshop, SED, University of Southern California, University of California, Los Angeles.

<sup>&</sup>lt;sup>†</sup>François Le Grand: emlyon business school, legrand@em-lyon.com; Xavier Ragot: SciencesPo-CNRS and OFCE, xavier.ragot@gmail.com. Corresponding author: Xavier Ragot.

if the default risk does not materialize, the discussion about its very possibility is a clear sign of the deterioration of the perceived safeness of public debt in some countries. The second feature affecting the global economy is the apparent shortage of safe assets. Observing the downward trend in the return on US public debt, Caballero and Farhi (2017), Barro and Mollerus (2016), and Hall (2016), among others, discuss the effects of a shortage of safe assets.<sup>1</sup> These two facts are obviously related, as sovereign default almost mechanically reduces the quantity of safe asset. In addition, the quantity of safe asset affects the level of the real interest rates and the default probability, through its effect on both the incentives to accumulate assets and on the opportunity cost to default.

The goal of this paper is to investigate the interactions between the quantity of safe assets and sovereign default in general equilibrium. We are interested in both positive and normative questions. What are the determinants of the quantity of safe assets when the sovereigns can choose to strategically default? Does the market economy generate too much safe asset, as a general equilibrium outcome? Should a world institution issue a safe asset? In which amount? This last question was, for instance, discussed in policy circles at the IMF before the decision to issue Special Drawing Rights.

To contribute to answer these questions, this paper presents a general equilibrium model of default, where sovereigns can borrow or lend on international markets, and possibly default on their debt, in order to smooth idiosyncratic income shocks. Elaborating on the Eaton and Gersovitz (1981) literature, the basic friction we consider is the lack of complete insurance market for country specific risk. Countries can only issue non-contingent claims, on which they can default. Equilibrium default can occur when the burden of the debt is high in bad times and when the opportunity cost of default is low. After the work of Aguiar and Gopinath (2006) and Arellano (2008) among others, this framework has become a benchmark to study sovereign default (see the recent exposition of Amador and Aguiar 2015). In this literature, the papers assume that the real interest rate is exogenous, so that they can investigate the amount of debt and the default decision in rich environments (see the literature review below). Our contribution is to endogenize the world interest rate by developing a tractable general equilibrium model.

We model the world economy as a continuum of countries borrowing and lending to each other to smooth idiosyncratic shocks. Countries can default on their debt depending on the

<sup>&</sup>lt;sup>1</sup>See also Gorton, Lewellen, and Metrick (2012) for a measure of the safe asset share in the US economy.

endogenous intertemporal cost of default. When countries default, they are temporarily excluded from financial markets. The world interest rate is determined by the international financial market clearing. An equilibrium in this economy is characterized by the default policy, the world wealth distribution and world interest rates. An alternative angle to consider this economy is to think about a Bewley economy where agents do not face exogenous credit constraint but can possibly default, if they find it optimal to do so. Surprisingly, the constrained-efficiency of market economies and of optimal policies in these environments have not been studied yet.<sup>2</sup> To do so, we use the strategy of Davila, Hong, Krusell, and Rios-Rull (2012), developed in incomplete-market economies without default, to derive the constrained-efficient allocation. We then analyze the properties of the market economy, and notably the welfare and quantity of safe asset, determining international risk sharing. We derive three main results.

The first result is that the interaction between incomplete markets and default generate multiple equilibria, differing in the quantity of the safe asset and interest rates. When the interest rate is high, the incentives to save for self-insurance are high. As countries save more, there is less default because the opportunity cost to default is high. Participating in financial markets is highly valued as risk sharing is important. A high supply of safe asset is consistent with a low default risk. On the contrary, with a smaller supply of safe assets available for self-insurance, there is more default in general equilibrium, as the relative gain to participating in financial markets is low. The multiple equilibria can be ranked according to their aggregate welfare. Equilibrium with less default unambiguously corresponds to higher welfare, compared to equilibrium with more defaults. The multiplicity of equilibria crucially thus depends on the endogeneity of the interest rate.

Second, we show that the market equilibrium with the highest welfare is still constrainedinefficient. The reason being that when markets are incomplete, prices do not convey the right incentives to issue safe assets. In other words, there is a pecuniary externality, implying a lack of risk sharing and a too small quantity of safe asset in the market economy. Sovereigns do not properly internalize in their net saving and default decisions the social benefits of their debt as a safe asset for other countries.

Third, the welfare in the constrained-efficient economy can be reached thanks to the intro-

 $<sup>^{2}</sup>$ Livshits, MacGee, and Tertilt (2007) and Chatterjee, Cordae, Nakajima and Rios-Rull (2007) have used these environments to compare the effect of different bankruptcy rules in quantitative setups. We instead simplify the environment to derive optimal policies in these environments.

duction of an international fund. This fund issues interest-bearing assets that are financed with the voluntarily contribution of member states. More precisely, the countries joining the fund receive a subsidy, while countries belonging to the fund pay a non state-contingent contribution to balance the budget of the fund. There is an equilibrium where all countries are either excluded from financial markets because they have defaulted or they all voluntarily contribute to the fund. In this case, there is full risk sharing among countries participating in financial markets, but default still occurs in general equilibrium. We provide a calibration of the model using both data on sovereign default and on "disaster event", to reproduce both the default probability and income fluctuations. The optimal quantity of assets issued by the fund represents around 4.1% of world GDP. The countries average contributions is equal to 0.22% of their own GDP, and the subsidy given by the the fund to joining countries is 3.49% of their GDP.

The fund and its asset supply are obviously reminiscent of the IMF Special Drawing Rights (SDRs). These SDRs were issued in the early 70s after a world discussion about the scarcity of safe stores of value (see Williamson 2009 for a short history). SDRs are interest-bearing assets, whose interest rate is weekly determined as the average interest rate on the money markets for a basket of currencies.<sup>3</sup> There are nevertheless two main differences between the assets issued by the fund in our model and SDRs. First, the interest rate on the asset issued by the fund in our model and SDRs. First, the interest rate on the asset issued by the fund in our model should be linked to the interest paid by government for long-maturities (which is the horizon for self-insurance) instead of the interest on money market as for SDRs. Second, the outstanding amounts differ substantially. For SDRs, the outstanding amount is smaller than 0.3% of the world GDP in 2016, while, with our calibration, the optimal issuance of world safe asset should correspond to 4.1% of world GDP.

This paper contributes to several strands of the literature. First, it belongs to the literature on the safe assets. This recent literature (Gorton, Lewellen, and Metrick 2012, Barro and Mollerus 2016, Hall 2016, and Caballero and Farhi 2017) studies the implication of the possible lack of safe assets. Another large literature investigates the determinants of the quantity of the world safe asset and of the interest rate (Mendoza and Quadrini 2009, Azzimonti, de Francesco, and Quadrini 2014). We introduce the interaction between sovereign default and the quantity of a safe asset in this literature. Recently, Farhi and Maggiori (2017) propose a model of the

<sup>&</sup>lt;sup>3</sup>See the IMF website for precise definition and the value of the interest rate on SDRs: http://www.imf.org/ external/np/fin/data/sdr\_ir.aspx.

international monetary system and study various market structures and frictions. We instead focus only on the optimality of the supply of the world safe asset. The gain of this strategy is to better understand market failures along this dimension and highlight new mechanisms, such as equilibrium multiplicity and the optimal size of a world fund.

Second, this paper belongs to the vast literature on sovereign default, which builds on Eaton and Gersovitz (1981) framework. This literature reproduces the empirical pattern of sovereign default in partial equilibrium (Aguiar and Gopinath 2006, Arellano 2008 among others, as well as the recent survey of Amador and Aguiar 2015). In this literature, our contribution is to investigate normative implications in general equilibrium and the possibility of equilibrium multiplicity.<sup>4</sup>

Our paper also belongs to the literature about default in general equilibrium. First, Zame (1993) and Dubey, Geanakoplos, and Shubik (2005) study in a two-period economy featuring incomplete markets the role of default, when default generates explicit costs. The literature on household default (Athreya 2002, Chatterjee, Cordae, Nakajima and Rios-Rull 2007, Livhists, MacGee and Tertilt 2007) has studied general equilibrium models. We focus on a simpler model to understand the distortions of the market economy, and to derive optimal policies.

Section 2 presents the environment. Section 3 solves for the market equilibrium and studies equilibrium multiplicity. Section 4 presents the constrained-efficient allocation. Section 5 shows that this allocation can be decentralized with a fund. Section 6 presents our numerical application and Section 7 concludes.

## 2 Environment

#### 2.1 Set-up

Time is discrete  $t = 1, ..., \infty$ . The world economy is modeled as a continuum of small open economies. Countries are distributed in each period t according to a uniform distribution G over a segment I of length 1.<sup>5</sup> Each economy has a risky production technology and a benevolent government maximizes the utility on behalf of a unit mass of identical consumers. Countries

<sup>&</sup>lt;sup>4</sup>Mendoza and Yue 2012 endogenize the cost of default using a model of trade credit.

<sup>&</sup>lt;sup>5</sup>This representation of the world economies follows Clarida (1990) or Bai and Zhang (2010). There is a literature about the applicability of the law of large number in continuum economies –see Feldman and Gilles (1985) and Green (1994) among others. We simply here assume that the law of large number applies.

face idiosyncratic production shocks, but there is no aggregate shock at the world level.

Each country has identical, additive, and time-separable preferences over streams of consumption  $(c_t)_{t\geq 0}$  and labor supply  $(l_t)_{t\geq 0}$ . The period utility function over consumption c and labor supply l is assumed to be quasi-linear: u(c) - l, where the consumption utility function u is increasing, differentiable, and concave. Quasi-linearity simplifies the equilibrium structure and preserves the essence of default mechanisms. Hansen (1985) and Rogerson (1988) provide a microfoundation for the linear disutility in labor by assuming that labor is indivisible and that the planner allocates labor within the country through lotteries. The utility function is used by Lagos and Wright (2005) in a matching environment and by Scheinkman and Weiss (1986) and Challe, LeGrand, and Ragot (2013) in an incomplete-market environment without default, among others. In each country, the government maximizes the intertemporal welfare  $\sum_{n=1}^{\infty} \beta^t (u(c_t) - l_t)$ , where  $\beta \in (0, 1)$  is the constant discount factor.

Countries face a productivity risk that can be neither avoided nor insured. The productivity status of a given country can be in one of two states, which are described as *productive* (state p) and *unproductive* (state u). When a country is productive, it has access to a linear production technology, which transforms l units of labor into l units of final goods. The country labor supply can in addition be freely adjusted in every period. When a country is unproductive, it is restricted to supply an amount  $\bar{l} < 1$  of labor. This restriction will ensure that both the period utility level and the marginal utility of consumption for a country in a productive state are smaller than those of a country in an unproductive state.

The productivity status follows a first-order Markov chain with transition matrix  $\Pi = \begin{bmatrix} \alpha & 1-\alpha \\ 1-\rho & \rho \end{bmatrix} \in (0,1)^4$ , where for example the probability  $1-\alpha$  is the probability to switch from state p in the current period to state u in the next one.

#### 2.2 Financial markets and default

As is standard in the literature on sovereign default, following Eaton and Gersovitz (1981) and Bulow and Rogoff (1989), international financial markets are assumed to be plagued by two frictions. The first one is the incompleteness of insurance-markets. Countries cannot trade assets contingent on their next period productive status, but can only trade one-period bonds. Second, countries can default on their debt, but at the cost of a temporary exclusion from financial markets. We assume that both the current productivity status and the asset volume for each country is observable when debt is sold or bought on international markets. If a country defaults on its debt, it is excluded from international markets and has to live in autarky. Even though excluded countries are prevented from saving or borrowing, they are affected by the productivity shock.

To make the model empirically relevant, we follow Arellano (2008) in introducing an additional cost of default. Countries in autarky have only access to a less efficient production technology, through which 1 unit of labor is only transformed into  $\varphi \in (0, 1]$  units of consumption. Excluded productive countries have in every period the probability  $\theta \in (0, 1)$  to reenter financial markets, while excluded unproductive agents cannot reenter financial markets. When reentering financial markers, the country is endowed with no financial wealth.

The debt is traded through a unit mass of competing risk-neutral financial intermediaries. Saving countries lend to financial intermediaries, whereas borrowing countries borrow from them. Financial intermediaries diversify their risk across countries and therefore act as devices pooling the idiosyncratic risk. As a consequence, in absence of aggregate risk, financial intermediaries can offer a risk-free rate to saving countries and will charge a credit risk premium to borrowers according to their default probability.

We denote as q(B', s) the price of a claim on one unit of next period good for a country who chooses an amount of asset B' and which has the production status  $s = \{p, u\}$ . If B' > 0, the country is saving in a safe asset of price q(B', s) = q. The real interest rate on the safe asset is simply  $1 + r \equiv \frac{1}{q}$ . If B' < 0, the country is borrowing. The assumption of perfectly diversified intermediaries implies that the price q(B', s) can be expressed as:

$$q(B',s) = q \times (1 - \delta(B',s)),$$

where  $\delta(B', s)$  is the default probability of a country choosing a debt B' while its current status is  $s \in \{p, u\}$ . Both the price of a safe asset q and the default probability  $\delta(B', s)$  will be general equilibrium endogenous outcomes. A country in state  $s \in \{p, u\}$ , endowed with the beginningof-period wealth B, and trading an amount of debt B', and supplying a labor quantity l, will have a consumption equal to:

$$c = l + B - q (B', s) B'.$$
(1)

If the country is in a productive state, it can then freely adjust the labor effort l. If unproductive, the country will supply the fixed amount  $l = \overline{l}$ .

Since the default probability  $\delta(B', s)$  is not necessarily continuous in the debt level B', agents will have to compare different debt levels corresponding to different prices. Furthermore, budget set may not be convex. There is no guarantee of equilibrium uniqueness in the general case, and we will show that it is not the case. However, although the problem is not convex, it can be written in a recursive form (see Stokey and Lucas 1989, Theorem 9.4), which will allow us to simplify the exposition and to derive necessary first-order conditions.

#### 2.3 Welfare functions and financial market clearing

We define  $V_s^o(B)$  as the value function of a country participating in financial market, starting the current period with the asset holding B, while being in the productivity status  $s \in \{p, u\}$ .<sup>6</sup> The superscript o denotes a participating country which has the *option* to default on B. The country decides then whether to default or repay its debts by choosing the option that is associated to the highest welfare. We denote  $V_s^c(B)$  the value function of a country deciding to repay the debt B, while its status is s. The superscript c indicates that the country decide to *continue* to honor its debts. Similarly, the value function  $V_s^d$  is the value function of the country deciding to default and is independent of the defaulted amount. The superscript d denotes the *default* decision. The value function  $V_s^o(B)$  is thus equal to the maximum between the value functions associated to debt repayment or to default. Formally:

$$\forall B, \forall s \in \{p, u\}, \quad V_s^o(B) = \max\{V_s^c(B), V_s^d\}.$$
(2)

Let us now turn to the expression of value functions  $V^d$  and  $V^c$ . Since defaulted countries face the probability  $\theta$  to possibly reenter financial markets when productive with a zero wealth, the value function associated to default can be expressed as:

$$V_p^d = \max_l \left( u\left(\varphi l\right) - l \right) + \beta \alpha \left(\theta V_p^o\left(0\right) + \left(1 - \theta\right) V_p^d\right) + \beta (1 - \alpha) V_u^d,\tag{3}$$

$$V_u^d = u\left(\varphi \bar{l}\right) - \bar{l} + \beta (1-\rho)\left(\theta V_p^o\left(0\right) + (1-\theta) V_p^d\right) + \beta \rho V_u^d.$$
(4)

<sup>6</sup>For the sake of simplicity, we follow the notations of Arellano (2008).

In these two equations; the country is excluded from financial markets, and is prevented from borrowing and saving. The whole production  $\varphi l$  or  $\varphi \bar{l}$  of the period is consumed within the period.

When the country participates in financial markets and chooses not to default in the current period, the value function  $V^c$  can be expressed as follows, for any debt endowment B:

$$V_{p}^{c}(B) = \max_{l,B'} u \left( l + B - q \left( B', s \right) B' \right) - l + \beta \left( \alpha V_{p}^{o}(B') + (1 - \alpha) V_{u}^{o}(B') \right),$$
(5)

$$V_{u}^{c}(B) = \max_{B'} u\left(\bar{l} + B - q(B', s)B'\right) - \bar{l} + \beta\left((1 - \rho)V_{p}^{o}(B') + \rho V_{u}^{o}(B')\right),$$
(6)

where we used the budget constraint (1) to substitute for the consumption expression. As the value functions  $V_s^c$  for s = p, u are increasing in the wealth amount B, we can define the quantities  $\overline{B}^s \in \mathbb{R} \cup \{-\infty\}, s = p, u$  as:

$$\overline{B}^s = \min_{B \in \mathbb{R}} \{ B | V_s^c(B) \ge V_s^d \}, s = p, u.$$
(7)

In words, the quantity  $\overline{B}^s$  is the state-contingent threshold such that a country endowed with a wealth level above the threshold will decide not to default.

We can finally provide the expression for the clearing of financial markets. Since assets are in zero net supply and competition among financial intermediaries imply a zero-profit condition, the value of all net asset demands sums up to zero. Recalling that countries are in every period distributed along the segment I with the distribution G, financial market clearing can be formalized as follows:

$$\int_{i\in I} q\left(B^{\prime i}, s^{i}\right) B^{\prime i} G\left(di\right) = 0,\tag{8}$$

where  $B'^i$  is the asset demand of country  $i \in I$  in the current period. We focus on the steady state equilibrium where the price of the safe asset q is constant. We can provide a more formal definition of the equilibrium.

**Definition 1 (Competitive equilibrium)** A competitive equilibrium is a price q, risk premia  $(\delta(B,s))_{s=u,p,B\in\mathbb{R}}$  and policy functions  $(B'(B,s))_{s=u,p,B\in\mathbb{R}}$ ,  $l(B,p)_{B\in\mathbb{R}}$  and the level of production in autarky l<sup>d</sup> such that: 1) for a given price q and given risk premia  $(\delta(B,s))_{s=u,p,B\in\mathbb{R}}$ , policy functions solve the program (2)-(6); 2) the risk premia are consistent with perfect competition and full diversification for financial intermediaries; 3) the price q is such that the financial

market clears, i.e., that equation (8) holds.

## 3 Market equilibrium

We can present the main properties of the market equilibrium. We start with the next proposition, which characterizes the equilibrium price schedule, and which is standard in this type of environment.

**Proposition 1 (Price schedule)** There exists two asset thresholds  $\overline{B}^e$  and  $\overline{B}^u$ ,  $\overline{B}^e < \overline{B}^u < 0$  such that:

$$q\left(B',p\right) = \begin{cases} q & \text{if } B' \ge \overline{B}^u, \\ \alpha q & \text{if } \overline{B}^e \le B' < \overline{B}^u, \\ 0 & \text{if } B' < \overline{B}^e, \end{cases} \text{ and } q\left(B',u\right) = \begin{cases} q & \text{if } B' \ge \overline{B}^u, \\ (1-\rho)q & \text{if } \overline{B}^e \le B' < \overline{B}^u, \\ 0 & \text{if } B' < \overline{B}^e. \end{cases}$$

Proposition 1 states that the price schedule is a step function with two thresholds  $\overline{B}^e$  and  $\overline{B}^u$ , where it can be shown that  $\overline{B}^e < \overline{B}^u$ . If a country borrows less than the smaller threshold  $\overline{B}^u$ , then it will not default in the next period, no matter its idiosyncratic shock. In this case, the price of its claim is the price of a safe asset q. If the country borrows strictly more than  $\overline{B}^u$  but less than a second threshold  $\overline{B}^e$ , it will default in case of a bad idiosyncratic shock, but it will not do so in case of a good idiosyncratic shock. As a consequence, the price of the claims is the price of the safe claim corrected by the probability to be productive in the next period, which is  $1 - \rho$  for unproductive countries, and  $\alpha$  for productive ones. Finally, for borrowed amounts that are strictly higher than  $\overline{B}^e$ , the country defaults no matter the next-period idiosyncratic shock, so that the price of the claim is 0. An implication of Proposition 1 is that productive countries never default. Indeed, if a country will then always default in the next period, implying that the price of its claim is 0, which is a contradiction as the country cannot borrow in that case.

#### 3.1 Euler equations and budget constraints

Quasi-linearity in the labor supply considerably simplifies the analysis of the equilibrium. To begin with, the first-order condition for the labor supply l of productive countries in (5) yields

 $u'(c_0) = 1$ , where  $c_0$  is the consumption of productive countries. In consequence, we obtain that the consumption level is  $c_0 = u'^{-1}(1)$ . Using the budget constraint (1), the value function (5) becomes:

$$V^{c}(B,p) = V^{c}(0,p) + B.$$
(9)

In consequence, the value function of productive countries participating in financial markets is linear in the beginning-of-period asset endowment. All productive non-defaulting countries have therefore the same marginal utility for consumption, which is independent of their beginningof-period endowment.

We now prove that the equilibrium is characterized by a sequence  $\{B_k\}_{k=0,...,\infty}$ , such that all productive countries participating in financial markets save  $B_0$  and participating countries who are unproductive for  $k \ge 1$  consecutive periods save the same amount  $B_k$ . In equilibrium, the saving (or borrowing) decision of agents thus simply depends on their idiosyncratic production history, which considerably simplifies aggregation.

We adopt a guess-and-verify strategy to simplify the exposition. Assume that countries default after being unproductive for  $D \leq \infty$  consecutive periods, where the case  $D = \infty$  corresponds to countries never defaulting. The first-order conditions for asset choices are:

$$q = \beta \left( \alpha + (1 - \alpha)u'(c_1) \right), \tag{10}$$

$$qu'(c_k) = \beta \left(1 - \rho + \rho u'(c_{k+1})\right), \quad k = 1, \dots, D - 2,$$
(11)

$$qu'(c_{D-1}) \ge \beta,\tag{12}$$

where  $c_k$ , k = 1, ..., D - 1 denotes the consumption level of a participating country who is unproductive for the k consecutive previous periods. The budget constraints (1) can now be expressed as:

$$B_{k-1} = c_k - \bar{l} + qB_k, \ k = 1, \dots, D-2,$$
(13)

$$B_{D-2} = c_{D-1} - l + q(1-\rho)B_{D-1},$$
(14)

We denote by  $l_{k0}$  the labor supply of productive countries who have been unproductive for the k previous periods and just turned productive. By extension,  $l_{00}$  denotes the labor supply of productive countries who were already productive in the previous period, and  $l_{-10}$  is the labor supply of participating productive countries who were excluded from financial markets in the previous period and who just reentered financial markets. For the sake of simplicity, the beginning-of-period wealth of these countries will also be assumed to be null:  $B_{-1} = 0$ . The budget constraint (1) implies:

$$l_{k0} = u'^{-1}(1) - B_k + qB_0, \ k = -1, 0, \dots, D - 1,$$
(15)

The intuition for such a simple equilibrium structure, is that all productive countries choose the same consumption level  $u'^{-1}(1)$  and the same asset holding  $B_0$  due to the quasi-linearity of the utility function (and the linearity of the value function). The labor supply of productive countries  $l_{k0}$  adjusts to compensate for the difference in their beginning-of-period wealth levels. This wealth level depends on country's productive status in the previous period and in particular on the number k of consecutive past periods of being unproductive (with k = 0 corresponding to a productive status and k = -1 to a default status).

**Remark 1 (Notation)** To lighten notation, we will consistently use below the notation that we introduced in this section: For a variable  $x, x_k$  is the quantity associated to a country unproductive for k = 1, ..., D - 1 periods, while  $x_{k0}$  denotes the quantity for a productive country unproductive for the k = -1, 0, ..., D - 1 previous periods. For instance,  $x_{20}$  concerns a productive country who was unproductive in the last two previous periods. The quantity associated to productive country who was already productive in the previous period is  $x_{00}$ , while  $x_{-10}$  is for a productive country who was excluded. When the past status does not matter for a productive country –for instance in the case of consumption level–, we simply use the notation  $x_0$ .

#### 3.2 Default conditions and limited-heterogeneity equilibrium.

Countries will default after being unproductive for exactly D consecutive periods if the following conditions hold: (i) paying back debt dominates the default option if countries unproductive for less that D-1 consecutive periods; (ii) unproductive countries for exactly D consecutive periods would be worse off paying back their debt than defaulting; (iii) productive countries who have been previously unproductive for k consecutive periods choose to pay back their debt contracted while unproductive and do not default. These conditions can be formalized as follows:

$$V_{u}^{c}(B_{k}) \ge V_{u}^{d}, \ 0 < k < D,$$
(16)

$$V_p^c(B_k) \ge V_p^d, \ 0 < k < D.$$
 (17)

We can now express the financial market clearing condition. First, denote  $n_k$  the population of participating countries being unproductive for k consecutive periods for  $1 \leq k \leq D$ . From transition probabilities, we have that  $n_k = \rho^k n_0$  for k = 1, ..., D - 1, where  $n_0$  denotes the mass of productive countries participating in financial markets. The explicit expression of these population share is relegated to the Appendix A to save some space. Using these notations and substituting for equilibrium population shares and prices, the financial market equilibrium (8) can be expressed as follows:

$$\sum_{k=0}^{D-2} n_k B_k + (1-\rho)n_{D-1} B_{D-1} = 0.$$
(18)

To summarize, the equilibrium is is thus characterized by: 1) a number of periods D before defaults, with possibly  $D = \infty$ , if default never occurs; 2) consumption levels  $(c_k)_{k=0,...,D-1}$ , asset demands  $(B_k)_{k=0,...,D-1}$  and labor efforts  $(l_{k0})_{k=-1,0,...,D-1}$  satisfying Euler equations (10)–(12), budget constraints (13)–(15), while conditions (16)–(17) hold; 2) a price for the safe asset q, enabling the market to clear and equation (18) to hold.

As we show in our quantitative exercise of Section 6, the value D is finite for standard calibrations. The economy is then populated by a finite number of different types of countries, where each type chooses the same consumption and wealth. More precisely, D-1 unproductive types participating in financial markets are characterized by the number k = 1, ..., D-1 of consecutive periods for which they are unproductive. Furthermore, D+1 productive types participate in financial markets and are characterized by their status in the previous period, which could either be: (i) unproductive, (ii) productive, or (iii) excluded from financial markets. As explained above, productive countries choose the same consumption and asset holding and only differ with respect to their labor effort. In addition to these 2D participating types, 2 additional types of countries are excluded from financial markets, and differ according to their productive status. On the whole, the economy is populated by 2D+2 different types of countries.

We can now compute the welfare for each type of country. Consistently with our previous

notation, we denote  $V_k$  the intertemporal welfare of a country participating in financial markets and unproductive for k = 1, ..., D - 1 periods, while  $V_{k0}$  denotes the intertemporal welfare of a productive country participating in financial markets, which was productive in the k =-1, ..., D - 1 previous periods. Formally, we have:

$$V_{k0} = u\left(u'^{-1}(1)\right) - l_{k0} + \beta\left(\alpha V_{00} + (1-\alpha)V_1\right), \ k = -1, \dots, D-1,$$
(19)

$$V_k = u(c_k) - \bar{l} + \beta \left( (1 - \rho) V_{k0} + \rho V_{k+1} \right), \ k = 1, \dots, D - 1.$$
(20)

Using expressions (19) and (20), participation conditions (16) and (17) become  $V_k \ge V_u^d$  and  $V_{k0} \ge V_p^d$  for  $k = 1, \dots, D-1$ .

The next proposition further characterizes the equilibrium structure.

**Proposition 2 (Limited-heterogeneity equilibrium characterization)** The price q verifies  $q \ge \beta$  and we have either:

- (a)  $D = \infty, q > \beta$  and  $\lim_{k \to \infty} B_k = -\frac{\overline{l}}{1-q}$ ; or
- (b)  $D < \infty$  and  $V_p^c(B_{D-1}) = V_p^d$ .

Proposition 2 states that two types of equilibrium can exist. The first one (a) is an equilibrium, in which default never occurs. The second one (b) is an equilibrium, in which the borrowing limit  $B_{D-1}$  for unproductive countries is determined by their default incentive when becoming productive next period. In this equilibrium, unproductive countries borrow more and more as they stay unproductive, until they reach an amount for which they would default in the following period in case they become productive. Indeed, a productive country, who starts the period with a large amount of debt would have to provide a significant labor effort to repay its debt and it may thus be better off defaulting.

Although characterizing possible equilibrium structure, Proposition 2 is silent about the possible equilibrium uniqueness. Section 3.3 below proves that multiple equilibria are possible.

Finally, we will compare the multiple equilibria using an utilitarian welfare criterion. Other criterion could be used, but this one seems to be the most natural in heterogeneous-agent economies, for reasons already discussed in Aiyagari (1994), and is thus widely used in the literature. Using the share of different types of countries, the expression of aggregate welfare  $W^a$  is simply:

$$W^{a} = \sum_{k=-1}^{D-1} n_{k0} V_{k0} + \sum_{k=1}^{D-1} n_{k} V_{k} + n_{p}^{d} V_{p}^{d} + n_{u}^{d} V_{u}^{d}.$$
 (21)

Aggregate welfare is the sum of the intertemporal welfare of countries participating in financial markets, being both productive and unproductive, and the intertemporal welfare of countries excluded from financial markets, who are both productive and unproductive.

#### 3.3 Multiple equilibria

The previous section has characterized the equilibrium structure in the general case. We now explain why non-trivial multiple equilibria can occur, with different numbers of periods before default (i.e., different values of D), different amounts of liquidity, and different prices for the safe asset (and thus different values for the real interest rate).

Countries self-insure themselves in good times by purchasing the safe asset. The quantity of safe assets amounts to the risk-adjusted quantity of debt in the economy. When the safe asset is scarce, its price q is high and the risk-free interest rate r is low. Self-insurance is expensive as the return on savings is low, and countries have low incentives to save in good times for self-insurance motives. In consequence, the demand of countries for the safe asset is low and consumption smoothing is poor, as it can be observed from first-order conditions of participating agents (10). Participating in financial markets is therefore not very attractive since risk sharing and consumption smoothing benefits are small. In these conditions, the opportunity cost to default is low, and the number of defaults in equilibrium is high, implying a small D in equilibrium. The quantity of "liquidity" available either for borrowing or lending is low. This is in turns consistent with the fact that liquidity is scarce and the price q of the safe asset is high.

On the opposite, when liquidity is abundant, the welfare gains of self-insurance are high, while the incentives to default are low. In equilibrium, countries default less and D is high in equilibrium.

To our knowledge, this identification of the possibility of equilibrium multiplicity in models with incomplete markets and default is new in the literature. The key mechanism is the endogeneity of the price of the safe assets. As a consequence, such multiple equilibria cannot occur in environments where the price of the safe asset is exogenous. Indeed, with exogenous prices, Auclert and Ronglie (2016) show that the equilibrium is unique under some general conditions, that are fulfilled here.

Our main proposition in this section is the ability to rank these multiple equilibria according to our welfare criterion.

**Proposition 3 (Ranking of multiple equilibria)** Multiple equilibria can be ranked according to their aggregate welfare. The higher the number D of consecutive periods before default, the higher the aggregate welfare. As a consequence, the competitive equilibrium that maximizes the aggregate welfare is the one with the largest admissible D.

The proof can be found in Appendix. Intuitively, the economy with a lower amount of default (and a higher D) is characterized by a higher quantity of liquidity and therefore more risk sharing, which is welfare improving.

# 4 Identifying distortions in the market economy: The constrainedefficient allocation

We now investigate the distortions in the market equilibrium corresponding to the highest welfare and the largest D. We prove that this equilibrium is not constrained-efficient. Indeed, even though this equilibrium is the market equilibrium featuring the highest quantity of liquidity, the liquidity remains insufficient for constraint-efficiency, because countries do not completely internalize the social costs of default.

To analyze these distortions more formally, we solve for the allocation of a quasi-planner facing the same risk structure as in the market economy, but who cannot transfer resources across countries and who cannot prevent countries from defaulting.<sup>7</sup> The quasi-planner decides the consumption level, the saving amount and the labor supply of all countries, while internalizing the financial market clearing condition. Our approach is similar to the one of Davila, Hong, Krusell, and Rios-Rull (2012), who define a notion of constrained-efficiency in an incomplete insurance-market environment without default. As a consequence, the individual budget constraint of each country must be fulfilled. In addition, since the quasi-planner cannot forbid

<sup>&</sup>lt;sup>7</sup>The term quasi-planner refers to a planner maximizing social welfare with country-specific budget constraints as, for instance, in Veracierto (2008) in a different environment. A general theory for the construction of a quasi-planner economy to identify distortions in incomplete-insurance market environment is presented in LeGrand and Ragot (2017). We apply it here to economies with default.

default, participation constraints of countries must also hold. The optimal allocation of the quasi-planner defines the constrained-efficient allocation, when the pecuniary externality is internalized. We show in Section 5 how this constrained-efficient allocation can be implemented by a simple market mechanism.

We now turn to the formal program of the quasi-planner. We use the same notations as in the market economy of Section 2, except that we add a tilde for the quasi-planner allocation. First, the planner decides upon the optimally the default decision  $\tilde{D}$ . The quasi-planner also chooses the consumption  $\tilde{c}_0$ , the net savings and  $\tilde{B}_0$ , and the labor supply  $\tilde{l}_{k0}$  of productive countries who have been unproductive for  $k = -1, 0, \ldots, \tilde{D}$  consecutive periods. The planner also chooses the consumption  $\tilde{c}_k$  and the savings  $\tilde{B}_k$  of unproductive countries for  $k = 1, 2, \ldots, \tilde{D}$ consecutive periods. The net wealth of countries re-entering financial markets is set to zero, as in the market economy:  $\tilde{B}_{-1} = 0$ . The quasi-planner cares about the instantaneous aggregate utility  $\tilde{U}$  expressed as

$$\tilde{U} = \sum_{k=-1}^{\tilde{D}-1} \tilde{n}_{k0} \left( u \left( \tilde{c}_0 \right) - \tilde{l}_{k0} \right) + \sum_{k=1}^{\tilde{D}-1} \tilde{n}_k \left( u \left( \tilde{c}_k \right) - \bar{l} \right) 
+ \tilde{n}_p^d \left( u \left( \tilde{c}_p^d \right) - \tilde{l}_p^d \right) + \tilde{n}_u^d \left( u \left( \tilde{c}_u^d \right) - \bar{l} \right),$$
(22)

where the tilda in population shares highlights their dependence in D. Formal expressions of population shares are the same as in the market economy, given in Appendix A. The program of the quasi-planner can be expressed recursively as follows

$$\tilde{W} = \max_{\substack{\{\tilde{q}, \tilde{D}, \tilde{c}_{p}^{d}, \tilde{l}_{p}^{d}, \tilde{c}_{u}^{d}\},\\\{\tilde{l}_{k0}\}_{-1 \le k < \tilde{D}-1}, \{\tilde{c}_{k}\}_{1 \le k < \tilde{D}-1}, \{\tilde{B}_{k}\}_{0 \le k < \tilde{D}-1}} \tilde{U} + \beta \tilde{W}',$$
(23)

subject to the set of country budget constraints, which are similar to (13)-(15) in the market economy:

$$\tilde{B}_{k-1} = \tilde{c}_k - \bar{l} + \tilde{q}\tilde{B}_k, \quad 1 \le k < \tilde{D} - 1,$$
(24)

$$\tilde{B}_{\tilde{D}-2} = \tilde{c}_{D-1} - \bar{l} + \tilde{q} (1-\rho) \tilde{B}_{\tilde{D}-1},$$
(25)

$$\tilde{c}_0 + \tilde{q}\tilde{B}_0 = \tilde{l}_{k0} + \tilde{B}_k, \quad -1 \le k \le \tilde{D} - 1,$$
(26)

$$\tilde{c}_p^d = \varphi \tilde{l}_p^d$$
, and  $\tilde{c}_u^d = \varphi \bar{l}$ , (27)

and subject to individual participation constraints (similar to (16) and (17) in the market economy):

$$\tilde{V}_{k0} \ge \tilde{V}_p^d, \quad -1 \le k < D, \tag{28}$$

$$\tilde{V}_k \ge \tilde{V}_u^d, \quad 1 \le k < D.$$
<sup>(29)</sup>

The quantities  $\tilde{V}_{k0}$  and  $\tilde{V}_k$  represent the intertemporal welfare of different country types, as in (19) and (20) for the market economy. The last constraint of the quasi-planner program is the financial market clearing:

$$\sum_{k=0}^{\tilde{D}-1} \tilde{n}_k \tilde{B}_k = 0.$$
(30)

The key-difference between the market economy and the constrained-efficient allocation, is that the quasi-planner internalizes the effect of saving and default decisions on the price of the safe asset, and thus on aggregate risk sharing.

The solution of the quasi-planner program is straightforward. We present here the corresponding first-order conditions to provide intuitions about the constrained-efficient allocation. We denote as  $\lambda$  the Lagrange multiplier of the financial market clearing constraint (30), which corresponds to the social value for the planner of one additional unit of world safe asset. Euler equations can be written as:

$$\tilde{q} + \lambda = \beta \left( \alpha + (1 - \alpha) u'(\tilde{c}_1) \right), \tag{31}$$

$$\tilde{q}u'(\tilde{c}_k) + \lambda = \beta \left(1 - \rho + \rho u'(\tilde{c}_{k+1})\right), \quad 1 \le k \le \tilde{D} - 2, \tag{32}$$

$$q\tilde{u'}\left(c_{\tilde{D}-1}\right) + \lambda \ge \beta. \tag{33}$$

Note that full risk sharing among participating countries is a solution of the quasi-planner program. Indeed, a solution to the set of equations (31)–(33) is  $\tilde{q} + \lambda = \beta$  and  $u'(\tilde{c}_k) = 1$  for  $k = 0, \ldots, \tilde{D} - 1$ . The quasi-planner distorts the saving choices of countries, so as to reach full risk sharing for countries participating in financial markets. This fosters the intertemporal welfare of participating countries, and therefore decreases the default incentives. This maximizes the quantity of the safe asset and aggregate welfare. The equilibrium value of  $\tilde{D}$  is pinned down by the participation constraints (28) and (29). Indeed, full risk sharing is obtained by countries accumulating debt when unproductive and continue to do so until they would be better off defaulting instead of paying back the debt when becoming productive. The price of the safe asset is then determined by the world financial market equilibrium together with all budget constraints. The next proposition summarizes this quasi-planner allocation.

**Proposition 4 (Constrained-efficient allocation)** The constrained-efficient allocation is characterized by  $\tilde{c}_k = u'^{-1}(1)$  for  $k = 1, ..., \tilde{D} - 1$ .

In general, the constrained-efficient allocation differs from the market allocation characterized in Section 3.1. To see this observe that  $\tilde{c}_k = u'^{-1}(1)$  in the market economy implies  $q = \beta$ . However, in this case, the saving decisions  $B_k$ ,  $k \leq \tilde{D} - 1$  have no reason to satisfy the financial market clearing. Since the quasi-planner internalizes the financial market clearing condition, this distorts Euler equations, and enables the quasi-planner to reach perfect risk sharing without the constraint of individual Euler equations.

This remark provides an intuition for the decentralization scheme presented in the next section. Indeed, perfect risk sharing implies an increase in the liquidity available for self-insurance in the market economy, but without distorting individual Euler equations. The decentralization scheme that we propose involves an international fund issuing liquidity and allowing countries to perfectly share risk.

# 5 A decentralization device: An international liquidity fund

We now show that an international liquidity fund can help to restore constrained-efficiency. The environment is similar to the market economy described in Section 2 (we keep the same notations), except the additional presence of the fund, issuing one-period risk-free debt financed by the contributions of member states.<sup>8</sup>

We assume that the contributions and the participation in the fund are regulated by the following rules.

1. The fund cannot implement contributions, conditional on the productive status of member countries.

<sup>&</sup>lt;sup>8</sup>The risk-free asset issued by the funds has some obvious connections with the Special Drawing Rights (SDRs) of the IMF. We discuss the differences below.

- 2. The fund can choose a different contribution for countries joining the fund in the current period.
- 3. Countries can freely choose to belong to the fund, and can always choose to opt out.
- 4. Contributing countries commit not to transact with countries not belonging to the fund.

Let us comment the above rules. First, we rule out contributions that can be contingent on the productive status and for instance be different for productive and unproductive countries. The fund could otherwise increase welfare using differentiated contributions, thereby providing partial insurance against the productive risk. We exclude this obvious comparative advantage for the fund, which could thus justify the existence of the fund. Instead, we assume that the fund is only able to discriminate countries just joining the fund from countries belonging to the fund for several periods (Assumption 2). This realistic weaker assumption is enough to exactly reproduce the constrained efficient aggregate welfare. Third, countries can decide to participate in the fund or not. The participation decision is sovereign. Fourth, contributing countries are prevented from transacting with non-contributing countries. Otherwise, countries would have no incentive to pay their contribution to the fund, as they could benefit from the world liquidity (and paying the world interest rate on any safe asset by absence of arbitrage) without paying contributing costs. No country would contribute to the fund.

We now turn to the formalization of the fund. We denote as F' the debt issued by the fund at the price q in a given period, while F is the amount the fund repays. Note that the fund pays the world risk-free interest rate, by absence of arbitrage. The total measure of countries participating in the fund is denoted  $m_p$ , while  $m_{-10}$  is the measure of countries just joining the fund in the current period. These quantities can differ from proportions  $n_p$  and  $n_{-10}$ . Countries joining the fund receive a transfer subsidy amounting denoted  $\tau_{-10}$ , and other contributing countries pay  $\tau$ . The notation  $\tau_{-10}$  refers to a country previously in the default state (state -1) becoming productive, re-entering financial markets and joining the fund with a zero wealth (state 0). With these notations, the fund budget constraint is:

$$F + m_{-10}\tau_{-10} = qF' + (m_p - m_{-10})\tau$$

The word financial market clearing condition can now be expressed as:

$$\sum_{k=0}^{D-1} n_k B_k = F'.$$
(34)

Finally, budget constraints of countries are the same as those in absence of fund in equations (13)-(15), except that countries additionally receive a transfer if they decide to join the fund. This contribution (or the subsidy) amounts to either  $\tau_{-10}$  or  $\tau$ , depending on whether the concerned country just joins the fund or not. The proposition summarizes the main result about the fund-economy.

**Proposition 5 (Equilibria in the family-head and decentralized economies)** There exist values for  $\tau$ ,  $\tau_{-10}$ , and F, such that the aggregate welfare in the fund-economy is identical to the one in a constrained-efficient economy. In the fund-economy, all countries either participate in the fund or are in autarky.

The proof can be found in Section D of Appendix. The equilibrium quantity F is determined to guarantee that the asset price is  $q = \beta$ . The transfers  $\tau$  and  $\tau_{-10}$  are set first to balance the fund budget, and second to implement the same default decision as in the constrained-efficient economy. At the equilibrium of the fund-economy, all countries participating to financial markets contribute to the fund.

It is noteworthy that Proposition 5 proves the existence of transfers  $\tau$ , and  $\tau_{-10}$ , and of a fund amount F for the aggregate welfare in the fund economy to be identical to the one in a constrained-efficient economy. However, the proposition is silent about uniqueness. Indeed, an equilibrium in which no country participates in the fund always exists. In that case, an atomistic country has no interest in contributing. The fund becomes meaningless and the economy is identical to the market economy.

This fund is obviously reminiscent of the IMF issuing Special Drawing Rights, which are international store of values. The SDR is a reserve asset issued by the IMF since 1970, with the explicit goal to reduce the world shortage of liquidity (see Williamson 2009 for a summary of the history of the introduction of SDRs). The main difference between the fund introduced in this section and the SDRs is that the interest on assets issued by the fund is a yearly interest rate, whereas the remuneration of SDRs is an average of short-run (3 months) interest rates on a basket of currencies.<sup>9</sup> The next section provides a quantification of the optimal amount of such a world liquidity.

## 6 A quantification of the size of the fund

We now provide a quantitative exercise to identify the distortions in the market economy and to quantify the size of the fund allowing to reach the constrained efficient allocation. To discipline the model we use both the literature on sovereign default, and the literature on disaster events, which is consistent with the two-state Markov structure we use for the idiosyncratic risk. The period is a year. The model has seven parameters, for which seven targets or values are provided.

The curvature of the utility function is set to  $\sigma = 2$ , as in Arellano (2008), and the discount factor is set to  $\beta = .96$  to obtain an annual world interest rate around of 4% in the world-market economy.

We set the probability of staying productive to 96.5% such that the probability to switch from productive to unproductive is 3.5%. This is equal to the annual probability to enter into a disaster state, as found by Barro and Ursua (2011). The probability to stay unproductive is set to 0.87% such that the average number of periods in a unproductive state amounts to 7.7 years. This value is close to the average duration of a disaster state, approximately equal to 7 years. The production when unproductive is set to  $\bar{l} = 0.99$ , which implies a cumulative output loss before default of 7%. This is consistent with Tomz and Wright (2013), who report that output is 8% below trend when default occurs. The probability to reenter the economy is set to  $\theta = 15\%$ , which generates an average length of financial market exclusion of 7 years, which roughly matches the 6.5 years reported by Tomz and Wright (2013). Finally the cost of default is set to 1.5% of output ( $\varphi = 0.985$ ). This value is chosen to trigger default after 7 consecutive periods in the unproductive state (as one possible equilibrium, see below), which is the average length for a disaster. This value implies that 16.4% of all countries are in default, which is close to the empirical value of 19% found by Tomz and Wright (2013). An additional outcome of this calibration is that the unconditional probability of default amounts to 1.0%. This value is slightly below standard empirical estimates, which vary between 1.8% and 2.2%. Table 1 summarizes the parameter values and targets.

<sup>&</sup>lt;sup>9</sup>The weekly interest rates on SDRs is provided at http://www.imf.org/external/np/fin/data/sdr\_ir.aspx,

Parameters	Values	Model outcomes and references			
Discount factor	$\beta=0.96$	Interest rate around 4%			
Utility function	$\sigma = 2$	Arellano (2008)			
Persistence of good state	$\alpha=96.5\%$	3.5% disaster probability			
		(Barro and Ursua 2011)			
Persistence of disaster	$\rho=87\%$	Average duration of 7.7 years			
		(Barro and Ursua 2011)			
Labor in disaster state	$\bar{l} = 0.99$	7% cumulative GDP fall when default			
		(Tomz and Wright 2013)			
Prob. to re-enter	$\theta = 15\%$	7 years of exclusion after default			
		(Tomz and Wright 2013)			
Default cost	$\varphi=0.985$	Default after 7 years in unproductive state			
		(Barro and Ursua 2011)			

Table 1: Calibration

#### 6.1 Market equilibria

The first outcome of the exercise is that multiple equilibria are pervasive. The previous calibration is consistent with an equilibrium where D = 7, i.e. where default occurs after 7 years in the unproductive states, but also with a range of equilibria with D varying from D = 2 to D = 7 (there is no equilibrium for D > 7). As was explained in the previous section, the source of equilibrium multiplicity is the endogeneity of the benefit to participate in financial markets, which depends on the default decision. In Table 2, we report for all equilibria some key statistics: interest rate, the fraction of countries in default and the unconditional default probability.

D		3	4	5	6	7
Interest rate (%)		4.13	4.13	4.13	4.13	4.13
Share of countries in default $(\%)$	30.4	27.0	23.9	21.1	18.6	16.4
Unconditional default probability (%)		1.82	1.6	1.3	1.2	1.0

Table 2: Equilibrium properties for various equilibria indexed by D

together with the explicit interest rate formula.

The equilibrium interest rate is lower in equilibrium where countries default less and it rapidly converges to the value of 4.13%. Note that the complete market interest rate is 4.17%. When D raises, both the supply and the demand for liquidity increases, as a smaller number of countries default and countries need to self-insure for a larger number of periods in unproductive state before defaulting. Consequently, the interest rate does not change a lot when D increases. In addition, and as can be expected, both the fraction of countries in default and the equilibrium default probability decrease with D.

As shown in Section 3.3, more risk sharing takes place in equilibria with larger D. We illustrate this aspect in Figure 1, where we plot the consumption profiles as a function of the number of consecutive unproductive priods, for different values of D. For sake of clarity, we only report D = 2, 4 and 7. When the country defaults and remains unproductive, its consumption is equal



Figure 1: Consumption profiles as a function of the number of consecutive unproductive periods, for various D

to the autaraky value  $\varphi \bar{l} = 0.9751$ , which is therefore the lowest consumption value. In addition, the path of consumption is decreasing with the number of consecutive period in unproductive state (before default), for all equilibria under consideration. Consumption-smoothing is thus imperfect. The implications for welfare are presented in Section 6.3 below.

#### 6.2 The equilibrium for the fund-economy

We now derive the equilibrium where the fund issues debt to replicate the constrained-efficient aggregate welfare. In the constrained-efficient equilibrium, we obtain  $D^{opt} = 10$ : countries default after being unproductive for 10 periods. The world interest rate is 4.17%, which is exactly the complete market interest rate  $1/\beta - 1$ . All countries participating in financial markets, either productive or not, enjoy the same consumption equal to 1. There is thus full risk sharing for countries participating in financial markets.

As shown in Section 5, this allocation can be implemented with three tools: a level of debt F, country-specific contributions  $\tau$ , and a subsidy for countries joining the fund,  $\tau_{-10}$ . With our calibration, implementing the constrained-efficient allocation implies the following fund characteristics. The fund must issue a debt level equal to 4.1% of world GDP. The average contributions of countries has to be equal to  $\tau = 0.22\%$  of their own GDP. Finally, the subsidy by the fund of countries re-entering financial markets and joining the fund is  $\tau_{-10} = 3.49\%$ . This amount is necessary to compensate for the net discounted value of expected contribution to the fund.<sup>10</sup>

#### 6.3 Welfare comparisons

We conclude our quantitative exercise by the comparison of the welfare properties of the multiple equilibria. For any equilibrium, we compute the aggregate welfare using the standard utilitarian criterion (see equation 21), which equally weights the intertemporal welfare of all countries. In Table 3, we report the aggregate welfare differences between market equilibria (for D varying from 2 to 7) and the fund economy where  $D^{opt} = 10$ . This aggregate welfare difference is expressed in consumption equivalent and corresponds to the relative decrease in consumption of each country, for the welfare in the fund economy to be identical to the one in the market

 $<sup>^{10} \</sup>rm Although$  relatively small, the size of the fund is much higher than the current amount of outstanding SDRs of the IMF which is less than 0.3% of world GDP in 2016.

economy for a given D.

D	2	3	4	5	6	7	
Welfare difference	-0.33	-0.27	-0.23	-0.18	-0.15	-0.11	
(in consumption equivalent $\%$ )							

Table 3: Welfare difference between market equilibria and the fund-economy in consumption equivalents.

As expected, the welfare in the fund-economy (where  $D^{opt} = 10$ ) is always higher than the welfare in the market economy, for any possible equilibrium. The welfare cost decreases with D. In other words, the higher D, the higher aggregate welfare. This is consistent with Proposition 3, illustrating that the higher D, the stronger the risk sharing.

To conclude, it is noteworthy that welfare differences reported in Table 3 are sizable. For instance, the welfare difference between the market equilibrium for D = 7 and the fund-economy amounts to 0.11% in consumption equivalent. This welfare difference is approximately equal to two or three times the welfare gains of eliminating business cycles in standard representativeagent analysis. The reason for this large welfare variations comes from the disaster state (being unproductive for seven periods), which is much more persistent than bad aggregate shocks in typical business cycle analysis.

# 7 Conclusion

We provide a tractable model where sovereign default can be studied in general equilibrium. This model allows us to derive normative properties of the equilibrium structure implied by the endogeneous quantity of the world safe asset. A salient property of the model is equilibrium multiplicity, that is due to an endogenous opportunity cost of default. In addition, all market equilibria exhibit too little risk sharing compared to a constrained efficient equilibrium. The additional result of the paper is that the constrained-efficient equilibrium can be attained by introducing a fund issuing safe asset based on the voluntary contribution of member states. The liability of the fund is around 4% of the world GDP. This quantity, though small, is still higher than the outstanding SDRs issued by the IMF.

# References

- AGUIAR, M., AND M. AMADOR (2015): "Chapter 11 Sovereign Debt," in Handbook of International Economics, ed. by E. Helpman, K. Rogoff, and G. Gopinath, vol. 4 of Handbook of International Economics, chap. 11, pp. 647–687. Elsevier.
- AGUIAR, M., AND G. GOPINATH (2006): "Defaultable Debt, Interest Rates and the Current Account," *Journal of International Economics*, 69(1), 64–83.
- AIYAGARI, R. S. (1994): "Uninsured Idiosyncratic Risk and Aggregate Saving," *Quarterly Journal of Economics*, 109(3), 659–684.
- ARELLANO, C. (2008): "Default Risk and Income Fluctuations in Emerging Economics," American Economic Review, 98(3), 690–712.
- ATHREYA, K. B. (2002): "Welfare Implications of the Bankruptcy Reform Act of 1999," Journal of Monetary Economics, 49(8), 1567–1595.
- AUCLERT, A., AND M. ROGNLIE (2016): "Unique Equilibrium in the Eaton-Gersovitz Model of Sovereign Debt," *Journal of Monetary Economics*, 84, 134–146.
- AZZIMONTI, M., E. DE FRANCESCO, AND V. QUADRINI (2014): "Financial Globalization, Inequality, and the Rising Public Debt," *American Economic Review*, 104(8), 2267–2302.
- BAI, Y., AND J. ZHANG (2010): "Solving the Feldstein-Horioka Puzzle With Financial Frictions," *Econometrica*, 78(2), 603–632.
- BARRO, R. J., AND A. MOLLERUS (2016): "Safe Assets," NBER Working Paper 20652, National Bureau of Economic Research.
- BARRO, R. J., AND J. F. URSUA (2011): "Rare Macroeconomic Disaster," NBER Working Paper 17328, National Bureau of Economic Research.
- BULOW, J. I., AND K. S. ROGOFF (1989): "Sovereign Debt: Is to Forgive to Forget?," American Economic Review, 79(1), 43–50.
- CABALLERO, R. J., AND E. FARHI (2017): "The Safety Trap," *Review of Economic Studies*, Forthcoming.
- CHALLE, E., F. LEGRAND, AND X. RAGOT (2013): "Incomplete Markets, Liquidation Risk and the Term Structure of Interest Rates," *Journal of Economic Theory*, 148(6), 2483–2519.
- CHATTERJEE, S., D. CORBAE, M. NAKAJIMA, AND J.-V. RÍOS-RULL (2007): "A Quantitative Theory of Unsecured Consumer Credit with Risk of Default," *Econometrica*, 75(6), 1525–1589.
- CLARIDA, R. (1990): "International Lending and Borrowing in a Staochastic Stationary Equilibrium," International Economic Review, 31(3), 543–558.
- DAVILA, J., J. HONG, P. KRUSELL, AND V. RIOS-RULL (2012): "Constrained Efficiency in the Neoclassical Growth Model With Uninsurable Idiosyncratic Shocks," *Econometrica*, 80(6), 2431–2467.
- DUBEY, P., J. GEANAKOPLOS, AND M. SHUBIK (2005): "Default and Punishment in General Equilibrium," *Econometrica*, 73(1), 1–37.

- EATON, J., AND M. GERSOVITZ (1981): "Debt with Potential Repudiation: Theoretical and Empirical Analysis," *Review of Economic Studies*, 48(2), 289–309.
- FARHI, E., AND M. MAGGIORI (2017): "A Model of the International Monetary System," Quarterly Journal of Economics, Forthcoming.
- FELDMAN, M., AND C. GILLES (1985): "An Expository Note on Individual Risk without Aggregate Uncertainty," *Journal of Economic Theory*, 35(1), 26–32.
- GORTON, G., S. LEWELLEN, AND A. METRICK (2012): "The Safe-Asset Share," American Economic Review, 102(3), 101–106.
- GREEN, E. (1994): "Individual-Level Randomness in a Nonatomic Population," Working Paper, University of Minnesota.
- HALL, R. E. (2016): "Understanding the Decline in the Safe Interest Rate," NBER Working Papers 22196, National Bureau of Economic Research.
- HANSEN, G. D. (1985): "Indivisible Labor and the Business Cycle," *Journal of Monetary Economics*, 16(3), 309–327.
- LAGOS, R., AND R. WRIGHT (2005): "A Unified Framework for Monetary Theory and Policy Analysis," *Journal of Political Economy*, 113(3), 463–484.
- LEGRAND, F., AND X. RAGOT (2017): "Optimal Fiscal Policy with Heterogeneous Agents and Aggregate Shocks," Working paper, SciencesPo.
- LIVSHITS, I., J. MACGEE, AND M. TERTILT (2007): "Consumer Bankruptcy: A Fresh Start," American Economic Review, 97(1), 402–418.
- MENDOZA, E., AND V. J.-V. R.-R. QUADRINI (2009): "Financial Integration, Financial Development and Global Imbalances," *Journal of Political Economy*, 117(3), 371–416.
- MENDOZA, E., AND V. Z. YUE (2012): "A General Equilibrium of Sovereign Default and Business Cycles," *Quarterly Journal of Economics*, 127(2), 889–946.
- ROGERSON, R. (1988): "Indivisible Labor, Lotteries and Equilibrium," Journal of Monetary Economics, 21(1), 3–16.
- SCHEINKMAN, J., AND L. WEISS (1986): "Borrowing Constraints and Aggregate Economic Activity," *Econometrica*, 54(1), 23–45.
- STOKEY, N. L., R. E. J. LUCAS, AND E. C. PRESCOTT (1989): Recursive Methods in Economic Dynamics. Harvard University Press.
- TOMZ, M., AND M. L. J. WRIGHT (2013): "Empirical Research on Sovereign Debt and Default," Annual Review of Economics, 5(1), 247–272.
- VERACIERTO, M. (2008): "Firing Costs and Business Cycle Fluctuations," International Economic Review, 49(1), 1–39.
- WILLIAMSON, J. (2009): "Understanding Special Drawing Rights (SDRs)," Policy Brief 09–11, Peterson Institute For International Economics.
- ZAME, W. R. (1993): "Efficiency and the Role of Default when Security Markets are Incomplete," *American Economic Review*, 83(5), 1142–1164.

# Appendix

# A Population shares

Using the model structure, one can deduce the following population shares

$$n_0 = \frac{\theta}{\theta + (1 - \theta)(1 - \alpha)\rho^{D - 1}} \frac{1 - \rho}{2 - \rho - \alpha},$$
(35)

$$n_k = \rho^{k-1}(1-\alpha)n_0 \text{ for } k = 1, \dots, D-1,$$
(36)

$$n_k = 0 \text{ for } k \ge D. \tag{37}$$

Denote  $n_u^d$  and  $n_p^d$  the mass of unproductive and productive agents who have defaulted and are excluded from financial markets:

$$n_p^d = \frac{(1-\theta)(1-\alpha)\rho^{D-1}}{\theta + (1-\theta)(1-\alpha)\rho^{D-1}} \frac{1-\rho}{2-\rho-\alpha},$$
(38)

$$n_u^d = \frac{(1 - (1 - \theta)\alpha)\,\rho^{D-1}}{\theta + (1 - \theta)\,(1 - \alpha)\rho^{D-1}} \frac{1 - \alpha}{2 - \rho - \alpha}.$$
(39)

With our timing convention, the number of excluded countries re-entering financial markets after having being excluded is then  $n_{-10} = \theta \left( \pi_{ee} n_p^d + \pi_{ue} n_u^d \right)$  in each period. It is then straightforward to deduce that the overall population (excluded and participating agents) of productive countries amounts to  $\frac{\pi_{up}}{\pi_{up} + \pi_{pu}}$  and this of unproductive countries to  $\frac{\pi_{pu}}{\pi_{up} + \pi_{pu}}$ . Note that these populations are independent of the quantity D.

## **B** Proof of Proposition 2

The proof is in three steps.

# **B.1** Proof that: $q \ge \beta$ .

Let us assume  $q < \beta$ . First, if  $D < \infty$ , equation (12), implies  $u'(c_{D-1}) \ge \beta/q > 1$ . We then deduce by induction from equation (11) that  $u'(c_k) > 1$  for k = 1, ..., D-1. Equation (10) then implies that  $q > \beta$ , which is a contradiction. Second, if  $D = \infty$ , we first show from equations (10) and (11) that  $1 > u'(c_k)$  for all k. Equation (11) also implies that  $\rho(u'(c_{k+1}) - u'(c_k)) <$   $(1-\rho)(u'(c_k)-1) < 0$  and the sequence  $(u'(c_{k+1}))_k$  is decreasing and positive. It thus admit a limit  $u'_{\infty}$ , which verifies from (11),  $u'_{\infty}\left(1-\frac{\beta}{q}\right) = \frac{\beta}{q}(1-\rho)(u'_{\infty}-1)$ . Since  $q < \beta$ , we must have  $u'_{\infty} > 1$ , which contradicts  $1 > u'(c_k)$  for all k.

### **B.2** Case $D = \infty$

Assume that  $q = \beta$ . Using the Euler conditions of all countries (10) and (11), one can show recursively that  $u'(c_k) = 1$  for all k. Hence,  $c_k = u'^{-1}(1)$ . Using the budget constraint one finds  $u'^{-1}(1) + \beta B_{k+1} = \overline{l} + B_k$ . It is then easy to show that  $\lim_{k\to\infty} B_k = -\infty$  (any other case being impossible due to financial market clearing condition). This is a contradiction from Proposition 1 as the there is a threshold for debt in any equilibrium under consideration. In consequence  $q > \beta$ .

Since  $q > \beta$ , we obtain from Euler equations (10)–(12):

$$1 - \alpha < (1 - \alpha)u'(c_1)$$
$$u'(c_k) - 1 < \rho(u'(c_{k+1}) - 1), \quad k \ge 1,$$

which implies

$$u'(c_k) - 1 \ge \rho^{-k} (u'(c_1) - 1)$$

with  $u'(c_1) > 1$ . In consequence,  $\lim_{k\to\infty} u'(c_k) = \infty$  and  $\lim_{k\to\infty} c_k = 0$ . Budget constraint (13) implies  $\lim_{k\to\infty} B_k = -\frac{\overline{l}}{1-q}$ .

#### **B.3** Case $D < \infty$

Since the default occurs at date D, we have  $V_k \ge V_u^d$  for  $k = 1, \ldots, D - 1$  and  $V_{k0} \ge V_p^d$  for  $k = -1, 0, \ldots, D - 1$ . Let assume that  $V_{D-1,0} > V_p^d$ . We have from equations (14) and (20):

$$V_{D-1} = u \left( B_{D-2} + \bar{l} - q B_{D-1} \right) - \bar{l} + \beta \left( (1-\rho) V_{D-1,0} + \rho V_u^d \right),$$

with  $V_{D-1} \ge V_u^d$  and  $V_{D-1}$  decreasing in  $B_{D-1}$ . In consequence, the agent u unproductive for D-1 periods could increase her welfare by decreasing slightly  $B_{D-1}$ . This would increase her welfare without affecting default incentives since  $V_{D-1,0} > V_p^d$ , which by continuity still holds after a small decrease in  $B_{D-1}$ .

# C Proof of Propositions 3

Let  $D \ge 1$ . Using equations (35)–(39) of Section A in Appendix, we can compute the impact of D on population shares.

•  $n_0$  goes up with D:

$$\frac{\partial n^p}{\partial \rho^{D-1}} = -\frac{(1-\theta)(1-\alpha)n^p}{\theta + (1-\theta)(1-\alpha)\rho^{D-1}} < 0.$$

- $n_p^d$  goes down with D since the total population of p agents remains constant.
- $\sum_{k=1}^{D-1} n_k$  goes up with D:

$$\frac{\partial n^u}{\partial \rho^{D-1}} = -\frac{(1-\theta)(1-\alpha)n^p}{\theta + (1-\theta)(1-\alpha)\rho^{D-1}} - \frac{\theta}{\theta + (1-\theta)(1-\alpha)\rho^{D-1}}\frac{1-\alpha}{2-\alpha-\rho} < 0.$$

•  $n_u^d$  goes down with D since the total population of u agents remains constant.

Consider the planner program that can rewritten as:

$$W = \max_{D \in \mathbb{N}_{*}} \max_{\substack{\{c_{p}^{d,D}, l_{p}^{d,D}, c_{u}^{d,D}\}, \\ \{l_{k0}^{D}\}_{-1 \le k < D}, \{c_{p}^{L}, B_{k}^{D}\}_{0 \le k < D} \in \mathcal{A}_{D}} U + \beta W',$$

where  $\mathcal{A}_D$  is the feasible set for a given  $D \in \mathbb{N}_*$ . We have added superscripts D to allocations in order to highlight the dependence in D. We have

$$U = \sum_{k=-1}^{D-1} n_{k0}^{D} \left( u \left( c_{0}^{D} \right) - l_{k0}^{D} \right) + \sum_{k=1}^{D-1} n_{k}^{D} \left( u \left( c_{k}^{D} \right) - \bar{l} \right) + n_{p}^{d,D} \left( u \left( c_{p}^{d,D} \right) - l_{p}^{d} \right) + n_{u}^{d,D} \left( u \left( c_{u}^{d,D} \right) - \bar{l} \right)$$

where we also added the D superscript to population proportions. Let us show that  $\mathcal{A}_D \subset \mathcal{A}_{D'}$ for any  $D' \geq D$ . To do so, we consider a feasible allocation  $\left(c_p^{d,D}, l_p^{d,D}, c_u^{d,D}, \{l_k^D\}_{-1 \leq k < D}, \{c_{p,k}^D, B_k^D\}_{0 \leq k < D}\right) \in \mathcal{A}_D$ . Using budget constraint (1), we can show that the following inequalities hold

$$B_{k-1}^{u,D} \ge c_k^{u,D} - \bar{l} + p B_k^{u,D}, 0 \le k < D - 1,$$
(40)

$$B_{D-2}^{u,D} \ge c_{D-1}^{u,D} - \bar{l} + p \left(1 - \rho\right) B_{D-1}^{u,D},\tag{41}$$

$$c_u^{d,D} \le \varphi \bar{l},\tag{42}$$

$$c_{p,k}^{p,D} + pB_0^{p,D} \le l_k^{p,D} + B_k^{p,D}, 0 \le k < D,$$
(43)

$$c_p^{d,D} \le \varphi l_p^{d,D}.\tag{44}$$

From the previous remark about the impact of D on population shares, we know that the population of p and u agents who are excluded is shrinking, while the population of p and u countries who participate is going up. We define for unproductive countries

$$\begin{split} B_k^{D'} &= \begin{cases} B_k^D & \text{for } n_k^D \text{ countries} \\ 0 & \text{for } n_k^{D'} - n_k^D \text{ countries} \\ \end{cases},\\ c_k^{D'} &= \begin{cases} c_{p,k}^D & \text{for } n_k^D \text{ countries} \\ c_{p,k}^d & \text{for } n_k^{D'} - n_k^D \text{ countries} \\ \end{cases},\\ l_k^{D'} &= \begin{cases} \bar{l} & \text{for } n_k^D \text{ countries} \\ \varphi \bar{l} & \text{for } n_k^{D'} - n_k^D \text{ countries} \\ \varphi \bar{l} & \text{for } n_k^{D'} - n_k^D \text{ countries} \\ \end{cases},\\ c_u^{d,D'} &= c_u^{d,D} \text{ for } n_u^{d,D'} \text{ countries},\\ l_u^{d,D'} &= \varphi \bar{l} \text{ for } n_u^{d,D'} \text{ countries}, \end{split}$$

where the idea is to "view" some excluded agents for D as participating agents for D', which is possible since we have (i)  $n_k^D \leq n_k^{D'}$ , (ii)  $n_u^{d,D'} \leq n_u^{d,D}$ , and (iii)  $n_k^D + n_u^{d,D} = n_k^{D'} + n_u^{d,D'}$ . It is obvious that constraint (42) holds for D', and that (40) and (41) also hold for D' (for a population of  $n_k^D$  agents that were in the same state for D). For the remaining population of  $n_k^{D'} - n_k^D$  agents, equation (40) for D' becomes  $c_u^{d,D'} - \bar{l} = c_u^{d,D} - \bar{l} + 0 = (\varphi - 1)\bar{l} \leq 0$ , which therefore holds. By the same token, equation (41) for D' also holds. For productive agents, we similarly define

$$\begin{split} B_0^{D'} &= \begin{cases} B_0^D & \text{for } n_0^D \text{ countries} \\ 0 & \text{for } n_0^{D'} - n_0^D \text{ countries} \end{cases}, \\ c_0^{D'} &= \begin{cases} c_0^D & \text{for } n_0^D \text{ countries} \\ c_p^{d,D} & \text{for } n_0^{D'} - n_0^D \text{ countries} \end{cases}, \\ l_{k0}^{D'} &= \begin{cases} l_{k0}^{p,D} & \text{for } n_{0k}^D \text{ countries} \\ \varphi l_p^{d,D} & \text{for } n_{0k}^D \text{ countries} \end{cases}, \\ c_p^{d,D'} &= c_p^{d,D} \text{ for } n_p^{d,D'} \text{ countries}, \\ l_p^{d,D'} &= \varphi l_p^{d,D} \text{ for } n_p^{d,D'} \text{ countries}, \end{cases}$$

Note that equation (44) holds for D' by construction. Regarding, equation (43), we have  $c_p^{d,D} \leq \varphi l_p^{d,D} \leq l_p^{d,D} \leq l_p^{d,D'}$ , which therefore holds.

In consequence the allocation  $\left(c_p^{d,D}, l_p^{d,D}, c_u^{d,D}, \{l_k^D\}_{-1 \le k < D}, \{c_k^D, B_k^D\}_{0 \le k < D}\right)$  belongs to the feasible set  $\mathcal{A}_{D'}$ . We deduce that  $\mathcal{A}_D \subset \mathcal{A}_{D'}$ .

This has two consequences:

- 1. for a given price, feasible sets are increasing in D (for the inclusion). This implies that any welfare level for D can be reached for any  $D' \ge D$ . The welfare is therefore increasing in D.
- by the same token, we can therefore rank competitive equilibria by an aggregate welfare criterion using D. This proves Lemma 3.

# D Proof of the proposition 5

To simplify the notations, we denote without variable and functions in the fund-economy with the same notations as in Sections 2 and 3. For instance, remind that a variable x related to a country re-entering financial markets is denoted  $x_{-10}$ . The program of such a country is

$$V_{-10} = u(c_{-10}) - l_{-10} + \beta \left( \alpha V_{00}(B_0) + (1 - \alpha) V_1(B_0) \right),$$
  
s.t.  $c_{-10} = l_{-10} + \tau_{-10} - qB_0.$ 

The budget constraint of the fund is

$$(m_p - m_{-10})\tau + qF = F + m_{-10}\tau_{-10}, \tag{45}$$

while the financial market clearing can be expressed as

$$\sum_{k=0}^{D-1} n_k B_k = F.$$
 (46)

An equilibrium for the fund economy is a default horizon D, a fund policy  $(F, \tau_{-10}, \tau)$ , an asset price q, allocations  $c_p^d, l_p^d, c_u^d, \{l_k\}_{-1 \le k < D}, \{c_k, B_k\}_{0 \le k < D}$ , such that 1) for a given price qand given fund policy  $(F, \tau_{-10}, \tau)$ , allocations are consistent with individual country programs; 2) the price q is such that the financial market clears, i.e., that equation (46) holds; 3) the fund policy is balanced : Equation (45) holds. Note that in this equilibrium, it is straghtforward to show that Proposition 2 holds, as the proof doesn't depend on contributions being paid or not by countries.

We start proving the following lemma.

**Lemma 1 (existence)** There exists  $(F, \tau, \tau_{-10})$ , such that the fund economy equilibrium is characterized by:  $D = \tilde{D}$ ,  $q = \beta$ , and all countries participating in financial markets participate in the fund.

We show how to find  $(F, \tau, \tau_{-10})$  satisfying the following conditions:

$$m_p = n_p \text{ and } m_{-10} = n_{-10},$$
(47)

$$q = \beta, \tag{48}$$

$$F = \frac{n_p - n_{-10}}{1 - \beta} \tau - \frac{n_{-10}}{1 - \beta} \tau_{-10}, \tag{49}$$

$$V_{-10}(0) = \tilde{V}_{-10}(0). \tag{50}$$

Condition (47) guarantees that all countries participating in financial markets participate in the fund. Condition (48) ensures that the price is the optimal one, such that the Euler equations are not distorted. Condition (49) ensures that the budget of the fund is balanced. Finally, (50) guarantees that  $D = \tilde{D}$ , as shown below.

#### Proof of lemma 1

The proof consists in proving that finding  $(F, \tau, \tau_{-10})$  such that equations (47)–(50) hold in the fund economy equilibrium boils down to solving a linear. We follow a guess and verify strategy. assume that equations (47)–(50) hold and that  $D = \tilde{D}$  and  $q = \beta$ . We derive four linear conditions defining  $(B_0, F, \tau, \tau_{-10})$ . Then for these values of  $(F, \tau, \tau_{-10})$ , the fund market economy equilibrium must satisfy  $D = \tilde{D}$  and  $q = \beta$ .

#### **First equation**

As  $V_{-10}(0) = \tilde{V}_{-10}(0)$ , the welfare in the fund-economy when reentering the economy is the same as in the constrained-efficient economy. It is easy to see that this implies that the autarky welfare levels in both economies are the same:  $V_p^d = \tilde{V}_p^d$  and  $V_u^d = \tilde{V}_u^d$ .

Using results of Proposition 2, we must have in the market economies  $V_p^d = V_c^p(B_{D-1})$ . This yields a first constraint on  $(F, \tau, \tau_{-10})$ .

Indeed, as  $D = \tilde{D}$  and  $q = \beta$ . The first order conditions, that are identical to (10)–(12), for  $k = 0, \ldots, D - 1$  yield

$$c_0 = c_k = u'^{-1}(1) \equiv c.$$

Any participating country is fully insured against the productivity risk. The budget constraints, similar to (24) and (25), of unproductive countries except that they include the payment of the fund fee  $\tau$  implies

$$B_{k+1} = \frac{\bar{l} - \tau - c}{\beta} + \frac{B_k}{\beta},$$

or after backward iteration:

$$B_k = \frac{B_0}{\beta^k} + (c - \bar{l} + \tau) \frac{1 - \beta^{-k}}{1 - \beta}.$$
(51)

Then welfare of participating countries that are unproductive for  $k \leq \tilde{D} - 1$  periods is

$$V_k = u(c) - \bar{l} + \beta \left( (1 - \rho) V_{k0}(B_k) + \rho V_{k+1} \right),$$
  
=  $u(c) - \bar{l} + \beta (1 - \rho) V_p^c(0) + \beta (1 - \rho) B_k + \beta \rho V_{k+1}$ 

Solving forward yields:

$$V_{1} = \left(u(c) - \bar{l} + \beta(1-\rho)V_{p}^{c}(0)\right) \frac{1 - (\beta\rho)^{D-1}}{1 - \beta\rho} + (c - \bar{l} + \tau)\frac{\beta(1-\rho)}{1-\beta} \frac{1 - (\beta\rho)^{D-1}}{1 - \beta\rho} + \left(B_{0} - \frac{c - \bar{l} + \tau}{1 - \beta}\right) (1 - \rho^{D-1}) + (\beta\rho)^{D-1}V_{u}^{d}.$$
(52)

The welfare of a productive participating country with a wealth 0 in the fund economy is

$$V_p^c(0) = u(c) - (c + \tau + \beta B_0) + \beta \alpha V_p^c(0) + \beta \alpha B_0 + \beta (1 - \alpha) V_1,$$

or after the expression (52) of  $V_1$ :

$$(1 - \beta \alpha) V_{p}^{c}(0) = u(c) - (c + \tau + \beta B_{0}) + \beta \alpha B_{0} + \left(u(c) - \bar{l} + \beta (1 - \rho) V_{p}^{c}(0)\right) \beta (1 - \alpha) \frac{1 - (\beta \rho)^{D-1}}{1 - \beta \rho} + (c - \bar{l} + \tau) \frac{\beta^{2} (1 - \alpha) (1 - \rho)}{1 - \beta} \frac{1 - (\beta \rho)^{D-1}}{1 - \beta \rho} + \beta (1 - \alpha) \left(B_{0} - \frac{c - \bar{l} + \tau}{1 - \beta}\right) (1 - \rho^{D-1}) + (\beta \rho)^{D-1} \beta (1 - \alpha) V_{u}^{d}.$$
(53)

As  $V_p^c(B_{D-1}) = V_e^d$  becomes  $V_p^c(0) = V_p^d - B_{D-1}$  or using (51):

$$V_p^c(0) = V_e^d - \beta^{-(D-1)} B_0 - \frac{1 - \beta^{-(D-1)}}{1 - \beta} (c - \bar{l} + \tau).$$
(54)

Substituting (54) into (53), we obtain:

$$\Psi_1 \tau = \Psi_2 - \Psi_3 B_0, \tag{55}$$

with

$$\begin{split} \Psi_{1} &= \beta^{2}(1-\alpha)(1-\rho)\frac{1-(\beta\rho)^{D-1}}{1-\beta\rho}\frac{\beta^{-(D-1)}}{1-\beta} \\ &+ \frac{1}{1-\beta}\left(\rho^{D-1}\beta(1-\alpha) - \beta^{-(D-1)}\left(1-\beta\alpha\right)\right), \\ \Psi_{2} &= -\left(u(c)-\bar{l}\right)\left(1+\beta(1-\alpha)\frac{1-(\beta\rho)^{D-1}}{1-\beta\rho}\right) \\ &- (c-\bar{l})\left(\beta^{2}(1-\alpha)(1-\rho)\frac{1-(\beta\rho)^{D-1}}{1-\beta\rho}\frac{\beta^{-(D-1)}}{1-\beta} \\ &+ \frac{1}{1-\beta}\left(\rho^{D-1}\beta(1-\alpha) - \beta^{-(D-1)}\left(1-\beta\alpha\right)\right)\right) \\ &\left(1-\beta\alpha-\beta^{2}(1-\alpha)(1-\rho)\frac{1-(\beta\rho)^{D-1}}{1-\beta\rho}\right)V_{e}^{d} - (\beta\rho)^{D-1}\beta(1-\alpha)V_{u}^{d}, \\ \Psi_{3} &= \left(1-\beta\alpha-\beta(1-\alpha)\frac{\beta(1-\rho)+(1-\beta)(\beta\rho)^{D-1}}{1-\beta\rho}\right)\beta^{-(D-1)}. \end{split}$$

#### Second equation

 $V_p^c(0)$  is the welfare of a participating country contributing to the fund. Since  $V_{-10}$  is the welfare of a country joining the fund and thus receiving a transfer  $\tau_{-10}$ . We thus have  $V_{-10}(0) = V_p^c(0) + \tau + \tau_{-10}$ , as countries joining the fund after default have 0 wealth and receive  $\tau_{-10}$  instead of paying  $\tau$ . As we have (by assumption):

$$V_{-10}(0) = \tilde{V}_{-10}(0),$$

we obtain using (54):

$$\tilde{V}_{-10}(0) - \tau_{-10} = V_e^d - \beta^{-(D-1)} B_0 - \frac{1 - \beta^{-(D-1)}}{1 - \beta} (c - \bar{l}) + \left(1 - \frac{1 - \beta^{-(D-1)}}{1 - \beta}\right) \tau,$$

which becomes with (55)

$$\tau_{-10} = -V_e^d + \tilde{V}_{-10}(0) + \frac{1 - \beta^{-(D-1)}}{1 - \beta}(c - \bar{l}) + \left(1 - \frac{1 - \beta^{-(D-1)}}{1 - \beta}\right)\tau + \beta B_0.$$
(56)

#### Third equation

Using the expression (51) of  $B_k$  in the financial market equilibrium (46), using the expression (36) of  $n_k$ , we obtain

$$\frac{F}{(1-\alpha)n_0} = \frac{B_0}{1-\alpha} + \sum_{k=1}^{D-1} \rho^{k-1} \left( \frac{B_0}{\beta^k} + (c-\bar{l}+\tau) \frac{1-\beta^{-k}}{1-\beta} \right)$$
$$= \frac{B_0}{1-\alpha} + \left( B_0 - \frac{c-\bar{l}+\tau}{1-\beta} \right) \frac{1-\rho^{D-1}\beta^{-(D-1)}}{\beta-\rho} + (c-\bar{l}+\tau) \frac{1-\rho^{D-1}}{(1-\beta)(1-\rho)}$$
$$= B_0 \left( \frac{1}{1-\alpha} + \frac{1-\rho^{D-1}\beta^{-(D-1)}}{\beta-\rho} \right) + \frac{c-\bar{l}+\tau}{1-\beta} \left( \frac{1-\rho^{D-1}}{1-\rho} - \frac{1-\rho^{D-1}\beta^{-(D-1)}}{\beta-\rho} \right).$$
(57)

#### Fourth equation

The fourth equation is the budget constraint of the fund (49):

$$(1-\beta)F = -n_{-10}\tau_{-10} + (n_p - n_{-10})\tau.$$
(58)

We thus have four linear equations (55), (56), (57), and (58) with four unknowns in  $B_0$ , F,  $\tau$ ,  $\tau_{-10}$ . Except particular cases (which would correspond to a non-invertible matrix for the linear system – which is a zero-measure set), there is always a solution. These four equations insure that there exist a fund economy equilibrium where countries default after  $D = \tilde{D}$  periods, where  $q = \beta$  and where the fund budget is balanced.

We now prove that the welfare in the fund economy is the same as in the constrained-optimal equilibrium, which is the next Lemma.

**Proposition 6 (existence)** If  $q = \beta$  and  $D = \tilde{D}$ , then the welfare in the fund-economy is the same as the welfare in the constrained efficient allocation.

As before, we denote with tilde the constraint efficient allocation, and without tilde the fund-economy allocation. The welfare in the constrained efficient allocation is simply, with  $c = u'^{-1}(1)$  and  $c^d = u'^{-1}(1/\varphi)$ 

$$\tilde{U} = \sum_{k=-1}^{\tilde{D}-1} \tilde{n}_{k0} \left( u\left(c\right) - \tilde{l}_{k0} \right) + \sum_{k=1}^{\tilde{D}-1} \tilde{n}_k \left( u\left(c\right) - \bar{l} \right) + \tilde{n}_p^d \left( u\left(c^d\right) - \frac{c^d}{\varphi} \right) + \tilde{n}_u^d \left( u\left(\varphi \bar{l}\right) - \bar{l} \right).$$

Thus

$$\tilde{U} = \tilde{n}_p u\left(c\right) - \sum_{k=-1}^{\tilde{D}-1} \tilde{n}_{k0} \tilde{l}_{k0} - \sum_{k=1}^{\tilde{D}-1} \tilde{n}_k \bar{l} + \tilde{n}_p^d \left(u\left(c^d\right) - \frac{c^d}{\varphi}\right) + \tilde{n}_u^d \left(u\left(\varphi \bar{l}\right) - \bar{l}\right).$$
(59)

The goods market equilibrium is

$$\tilde{n}_p c = \sum_{k=-1}^{\tilde{D}-1} \tilde{n}_{k0} \tilde{l}_{k0}^p + \left(\sum_{k=1}^{\tilde{D}-1} \tilde{n}_k\right) \bar{l}.$$
(60)

In the fund-economy, as  $q = \beta$ , the consumption of all participating countries is c (whereas non-participating countries live in autarky). The welfare is

$$U = n_p u(c) - \sum_{k=-1}^{D-1} n_{k0} l_{k0} - \sum_{k=1}^{D-1} n_k \bar{l} + n_p^d \left( u(c^d) - \frac{c^d}{\varphi} \right) + n_u^d \left( u(\varphi \bar{l}) - \bar{l} \right).$$
(61)

The goods market equilibrium is

$$n_p c = \sum_{k=-1}^{D-1} n_{k0} l_{k0} + \left(\sum_{k=1}^{D-1} n_k\right) \bar{l}.$$
 (62)

As  $D = \tilde{D}$ , we have  $n_p = \tilde{n}_p$ ,  $n_{k0} = \tilde{n}_{k0}$ ,  $n_k = \tilde{n}_k$ ,  $n_p^d = \tilde{n}_p^d$  and  $n_u^d = \tilde{n}_u^d$ . In consequence, (60) and (62) imply that  $\sum_{k=-1}^{D-1} n_{k0} l_{k0} = \sum_{k=-1}^{\tilde{D}-1} \tilde{n}_{k0} \tilde{l}_{k0}^p$ . We deduce from (59) and (61) that  $U = \tilde{U}$ .